Author's response to reviews

Title: Comparison between minimally invasive, percutaneous osteosynthesis and locking plate osteosynthesis in 3- and 4-part proximal humerus fractures.

Authors:

Reinhold Ortmaier (r.ortmaier@gmail.com)
Verena Filzmaier (Verenafilzmaier@hotmail.com)
Wolfgang Hitzl (wolfgang.hitzl@pmu.ac.at)
Robert Bogner (r.bogner@salk.at)
Thomas Neubauer (thneubauer@aon.at)
Herbert Resch (h.resch@salk.at)
Alexander Auffarth (a.auffarth@salk.at)

Version: 2
Date: 20 May 2015

Author's response to reviews:

1. Added name of ethic committee line 128
2. Added a list of abbreviations line 363
3. Added acknowledgements line 380
4. Added line and page number